



Limnology and Oceanography Letters 8, 2023, 751–759 © 2023 The Authors. Limnology and Oceanography Letters published by Wiley Periodicals LLC on behalf of Association for the Sciences of Limnology and Oceanography. doi: 10.1002/10/21.0333

## LETTER

# Variation in fatty acid content among benthic invertebrates in a seasonally driven system

Per Hedberg,<sup>1</sup>\* Danny C. P. Lau,<sup>2,3</sup> Séréna Albert<sup>1</sup>,<sup>1</sup> Monika Winder<sup>1</sup>\*

<sup>1</sup>Department of Ecology, Environment and Plant Sciences, Stockholm University, Stockholm, Sweden; <sup>2</sup>Department of Aquatic Sciences and Assessment, Swedish University of Agricultural Sciences, Uppsala, Sweden; <sup>3</sup>Department of Ecology and Environmental Science, Umeå University, Umeå, Sweden

### Scientific Significance Statement

Algae produce essential nutrients, such as fatty acids (FAs) that are transported to animals living on the seafloor and fish that feed on them. With varying algal species, it is expected that the FA content of animals is changing, but this has not been tested with seafloor animals. Contrary to this, we find low seasonal variation in FAs in seafloor animals in a system with nutritious spring bloom and summer algae with low essential FAs. But we find differences in FAs between benthic animals that vary with trophic levels. This suggests that seafloor animals have evolved mechanisms to adapt to low-quality food inputs and that species composition of seafloor animals is important for determining food quality for fish.

#### **Abstract**

At temperate latitudes where seasonal changing environmental conditions strongly affect the magnitude, duration and species composition of pelagic primary production, macrobenthic organisms living below the photic zone rely on the sedimentation of organic matter as their primary energy source. The succession from nutritious spring blooms to summer cyanobacteria is assumed to reduce food quality for benthic primary consumers and their fatty acid (FA) profiles. In contrast, we find low seasonal variability in FA content of five benthic macroinvertebrates spanning two trophic levels in the Baltic Sea, a system with high seasonal variation in phytoplankton species composition. However, levels of the major FA groups vary greatly between benthic species. The results suggest that benthic macroinvertebrates have evolved FA metabolism adapted to degraded sedimenting material. Moreover, our study shows that species composition of benthic macrofauna rather than seasonal changing conditions affect availability of essential nutrients to higher trophic levels.

Benthic animals living below the photic zone rely on the sedimentation of organic matter (OM) produced by phytoplankton as the primary energy source (Kortsch et al. 2015; Griffiths et al. 2017). Benthic macroinvertebrates

\*Correspondence: per.hedberg@helsinki.fi; monika.winder@su.se

#### Associate editor: Sarah Fawcett

Author Contribution Statement: PH and MW designed the study. PH and SA collected samples from the field. PH prepared samples for FA analysis. DL together with PH analyzed samples for FAs. PH and MW analyzed and visualized the data. PH and MW wrote the manuscript. All authors contributed to the written work in discussions and editing.

Data Available Statement: Data are available in the DRYAD repository at https://doi.org/doi:10.5061/dryad.djh9w0w3r.

Additional Supporting Information may be found in the online version of this article.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

are themselves an important trophic vector and source of essential compounds produced by phytoplankton for fish, birds, and mammals (Benke and Huryn 2010). Benthic invertebrates often respond in biomolecule composition to inputs of sedimenting OM, which is, however, variable between species (Graf et al. 1982; Lehtonen and Andersin 1998; Lahdes et al. 2010). At temperate latitudes, seasonal changing environmental conditions strongly affect the magnitude, duration, and species composition of phytoplankton blooms (Lutz et al. 2007; Winder and Cloern 2010), and consequently sedimenting input for benthic animals. Spring blooms, largely composed of diatoms and dinoflagellates, are thought to be the major annual input of high-quality food, while summertime exports are more variable depending on species composition (Simis et al. 2017). Coastal ecosystems, including the Baltic Sea, are experiencing an increase in summer cyanobacteria due to eutrophication and climate warming, which is expected to decrease inputs of essential macromolecules and their availability for higher trophic levels (Lehtonen and Andersin 1998; Hjerne et al. 2019; Schmidt et al. 2020). However, little is known about dietary transfer of essential compounds in benthic consumers and how they respond to seasonally varying inputs of OM.

Phytoplankton are assumed to be the main producers of essential biomolecules, like amino acids and fatty acids (FAs) for consumers in general and benthic invertebrates residing in the nonphotic zone in particular (Petersen and Curtis 1980). FA composition varies between major phytoplankton taxa, with diatoms and dinoflagellates being rich in the essential polyunsaturated fatty acids (EFAs), including arachidonic acid (ARA, 20:4w6), eicosapentaenoic acid (EPA, 20:5w3), and docasahexaenoic acid (DHA, 22:6ω3), while cyanobacteria to a large extent are lacking these compounds (Galloway and Winder 2015). EFAs are generally not synthesized de novo by consumers and mainly assimilated from diet (Parrish 2008); however, there is increasing evidence for potential EFA synthesis in a wide variety of invertebrate animals (Kabeva et al. 2018). EFAs are important for fish nutrition and commonly linked to consumers' growth and fecundity (Brett and Müller-Navarra 1997). Consequently, shifts in phytoplankton species composition and particularly declines in nutritious spring blooms and increase in low-quality phytoplankton during summer periods is assumed to result in decreased EFAs at higher trophic levels, affecting ecosystem productivity (Taipale et al. 2016; Schmidt et al. 2020; Svedäng et al. 2022). Although EFA transfer pathways in aquatic environments are well documented within the pelagic food web (Brett and Müller-Navarra 1997; Arts et al. 2001), this field is largely understudied in benthic primary consumers (Monroig et al. 2013).

To address this gap, we study effects of seasonal changing phytoplankton species on FA quantity and composition of pelagic sedimenting OM and benthic macroinvertebrates. This study was performed in the Baltic Sea, where diatoms and dinoflagellates dominate the spring bloom and cyanobacteria the summer bloom (Fig. 1A). We explore this by analyzing seasonal FA content per dry weight, tracing of food sources' origin and partitioning of trophic levels using stable isotopes (SI) in OM and dominating benthic primary consumers.

#### **Methods**

Seston, sediment, and benthic animals were sampled at a coastal site in the north-western Baltic Proper (58°48'39.5"N, 17°36′25.1″E). The site was situated at 40 m depth with a bottom water salinity varying from 6 to 7 and temperature from 0.18°C to 6.84°C during our sampling occasions. Sampling was done in July, August, October, November 2018, and March, April 2019, representing summer, autumn, and spring bloom conditions. Seston was collected as a vertical integrated water sample from the surface to 20 m depth and was prefiltered through a 100- $\mu$ m mesh, then filtered onto GF/F filters and frozen at -80°C. Sediment was sampled with a boxcorer, subsampled with plexiglass tubes, after which the upper sediment (ca 9 mm) was sliced off, homogenized, and frozen at  $-80^{\circ}$ C. Animals were sampled with a benthic sled. We selected adult stages of common macrofauna species, including the surface and subsurface feeding amphipods, Monoporeia affinis and Pontoporeia femorata, respectively, the facultative suspension and deposit-feeding clam Macoma balthica, the deposit-feeding polychaete Marenzelleria spp. and the predatory priapulid worm Halicryptus spinulosus (referred hereafter to their genera). Animals were sieved from the sediment, placed in aerated filtered sea water for 24 h in order to facilitate gut evacuation, keeping Halicryptus separate. Macoma was separated from their shell and all animals were frozen at -80°C. For FA and SI analyses, samples were freeze-dried at  $-120^{\circ}$ C (one replicate for seston and sediment, three for each species) and pulverized. Further details of sample collection, and FA and SI analyses are provided in the Supporting Information. FAs were grouped into major groups: Saturated FAs (SAFAs), monounsaturated FAs (MUFAs), polyunsaturated FAs (PUFAs, including EFAs and their precursors alpha-linolenic acid [ALA] and linoleic acid [LIN]), and bacterial FAs (BAFAs); a full list of individual FAs analyzed is provided in Supporting Information Table S1.

Differences of FA group and EFA content between species for the whole sampling period and within species over time were analyzed with permutational multivariate ANOVA (PERMANOVA) using the vegan (Oksanen et al. 2020) and pairwiseAdonis (Martinez Arbizu 2020) packages after testing for normality and variances. Nonmetric Multidimensional Scaling (NMDS) with Bray–Curtis dissimilarity distance was used to explore similarity of individual FAs among benthic species, seston, and sediment based on percentages over the whole sampling period. Individual differences within certain groups of FAs, for each species over time and between species Hedberg et al.

23782242, 2023, 5, Downloaded from https://aslopubs

com/doi/10.1002/lol2.10333 by Swedish University Of Agricultural Sciences, Wiley Online Library on [11/10/2023]. See the Term

Wiley Online Library for rules of use; OA

articles

are governed by the applicable Creative Commons License



Fig. 1. Seasonal dynamics of phytoplankton as well as seston and sediment FA content in the Baltic Sea from July 2018 to April 2019. (A) Average monthly biovolume of major phytoplankton taxa. Note, ciliates constitute the mixotroph Mesodinium rubrum. X-axis abbreviations are month of the year. (B) Seston and sediment (upper 9 mm) content of major FA groups, including SAFAs, MUFAs, PUFAs, and BAFAs. SAFAs of seston is shown in the insert. (C) Seston and sediment EFAs, including ARA, DHA, and EPA. Data on plankton biovolume are available from the Swedish Meteorological and Hydrological Institute (SMHI) website (https://sharkweb.smhi.se/).

over time, SI data and C:N ratio were analyzed either with a one-way or two-way ANOVA in a generalized least squares model (gls) in the nmle package (Pinheiro et al. 2006). Relationships between major phytoplankton groups  $(mm^3 L^{-1})$ and individual FAs  $(mg g^{-1})$  in seston, sediment and animals were investigated with multiple linear regression models, on log-transformed FA data to fulfill assumptions of linearity. All analyses were done using R (R Core Team, 2020).

#### Results

Seston contained high levels of SAFA, ranging from 80 to about 400 mg  $g^{-1}$  or more than 90% of total FA, with all major FA groups and EFA peaking in October and November (Fig. 1B,C; Supporting Information Fig. S1). Seston C:N ratio reached up to 34 during the spring bloom (March-April) and varied between 5 and 16 for the remaining months, while sediment C:N was around 7.8 (Supporting Information Fig. S2).

23782242, 2023, S, Downloaded from https://aslopubs.onlinelibrary.wiley.com/doi/10.1002/102.10333 by Swedish University Of Agricultural Sciences, Wiley Online Library on [11/10/2023]. See the Terms and Conditions (https://aslopubs.onlinelibrary.wiley.com/doi/10.1002/102.10333 by Swedish University Of Agricultural Sciences, Wiley Online Library on [11/10/2023].

-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

In comparison to seston, sediment contents of major FA groups and individual EFA were orders of magnitude lower, not exceeding 0.6 mg g<sup>-1</sup> (Fig. 1B,C) and consisting mainly of the SAFA palmitic acid (16:0; Supporting Information Table S1). Sediment FA showed low seasonal variation, except for an increase of EPA and DHA in April (Fig. 1C).

Seasonal variation of major FA groups and individual EFAs in benthic species were unrelated to the succession in phytoplankton biomass of major taxa (p > 0.05) and showed generally low variation throughout the season, with two exceptions (Fig. 2; Supporting Information Fig. S1; Table S2). In *Marenzelleria*, total FAs differed between November (higher) and March (lower) at p < 0.05 (Fig. 2A; Supporting Information Table S3). *Monoporeia* had a slight increase of BAFAs in March before the spring bloom (p < 0.05) and a low C:N ratio of 5.7 in April (Fig. 2A; Supporting Information Table S4), while the  $\omega 3:\omega 6$  ratio was low during the summer and increased after the autumn and spring bloom (Supporting Information Fig. S3; Table S5).

In contrast to low seasonal changes, we observed variation among the benthic animals. The amphipods Monoporeia  $(29.9 \text{ mg g}^{-1})$  and *Pontoporeia*  $(35.1 \text{ mg g}^{-1})$  and the polychaete *Marenzelleria* (30.8 mg  $g^{-1}$ ) reached the highest total FA content averaged over the whole sampling period among all macroinvertebrates, which is largely due to high content of SAFAs and MUFAs in amphipods, and also PUFAs in Marenzelleria (Fig. 2A). In comparison, total FAs, SAFAs, and MUFAs were significantly lower for Macoma and Halicryptus (p < 0.001; Supporting Information Table S6). Marenzelleria had about two to three times as high levels of BAFAs than the other species (p < 0.01; Supporting Information Table S8), except for July and March (Fig. 2A; Supporting Information Table S8). BAFA values were similar for the amphipod species, except for a peak in March for *Monoporeia* (p < 0.05; Supporting Information Table S4), and low in Halicryptus and Macoma. Marenzelleria had a lower  $\omega 3:\omega 6$  ratio (less than 1) than all other species throughout the sampling period (p < 0.001; Supporting Information Table S5), indicative of high amounts of the 20:2w6 FA intake (Supporting Information Fig. S3; Table S1). In addition, Marenzelleria and Halicryptus had the lowest C:N ratio of about 5.2 with other species reaching up to 11.7 (Fig. 2A).

*Marenzelleria* reached the highest PUFA levels with an average of about 7.9 mg g<sup>-1</sup> related to high EPA and DHA values, while PUFA contents of all other species were significantly lower and ranged between 0.6 and 1.9 mg g<sup>-1</sup> throughout the sampling period (p < 0.01; Figs. 2, 3A; Supporting Information Table S7). DHA content in *Monoporeia* was significantly higher from November to April compared to August (Supporting Information Table S2). *Macoma* and *Halicryptus* had the lowest EFA content among the species and sediment contained minimal levels of EFAs (Fig. 3A).

The NMDS revealed separate clustering of sediment and seston from the animal species, being mostly associated with

SAFAs (Fig. 3B). Between species, the two amphipods and the clam displayed similar patterns and were positively associated to ALA, SAFAs, and other FAs. Likewise, *Marenzelleria* and *Halicryptus* clustered, positively associated with PUFAs like LIN, ARA, EPA, and DHA.

#### Stable isotopes

Seston  $\delta^{13}$ C and  $\delta^{15}$ N varied throughout the sampling period (Supporting Information Fig. S4), while SI values of sediments did not change (Fig. 4). Among all benthic macroinvertebrates (with the exception of April), *Halicryptus* had the highest trophic position according to its  $\delta^{15}$ N mean value of 11.02‰ (p < 0.05), followed by *Marenzelleria* with a mean of 9.32‰ (p < 0.01), while *Monoporeia*, *Pontoporeia*, and *Macoma* had similar  $\delta^{15}$ N value ranging from 7.21‰ to 7.74‰ (p > 0.1; Fig. 4; Supporting Information Table S9). All benthic invertebrates had similar  $\delta^{13}$ C values (-21.5) with higher values than sediment (-23.0) and seston (-26.5). *Monoporeia* was more negative in  $\delta^{13}$ C in April compared to all other months (p < 0.001; Supporting Information Table S10).

#### Discussion

We find that the most dominant benthic invertebrates show low seasonal variation in FA content in a system with large variation in phytoplankton species composition of different nutritional quality. Although seston and sediment FA values partially reflect seasonal change in phytoplankton production and sedimentation, these subtle changes were only translated to a few benthic macrofauna. By measuring quantitative FA content per dry weight, we find that EFA contents of benthic macrofauna are orders of magnitude higher compared to seston and sediment, and vary across benthic species. These findings indicate that by rather converting dietary EFA, the FA content of benthic species is regulated by diverse mechanisms.

Seston FA content remained low during the spring and summer period but increased in autumn (October-November), likely related to a phytoplankton bloom, zooplankton activity through the production of fecal pellets and particle resuspension coupled to the break-down of the stratified water column (Tiselius et al. 2012). In comparison to seston, sediment FA content, C:N ratio, and isotopic signatures were stable throughout the sampling period, except in April when EPA and DHA increased slightly. This increase is most likely linked to the settling of the spring phytoplankton bloom. Manifold lower sediment content, mainly consisting of the SAFA palmitic acid (16:0) suggests breakdown of FAs during sedimentation and low nutritional value of sediment, in accordance to previous findings (Goedkoop et al. 2000). These data are in line with other studies, indicating that spring blooms contribute to the highest level of fresh OM sedimentation (Spilling et al. 2014). Subsequent blooms throughout the vear are subjected to increased degradation, grazing by zooplankton and are generally lower in organic quantity and



Fig. 2. Seasonal dynamics of FA content and C:N ratio of benthic macroinvertebrate species in the Baltic Sea from July 2018 to April 2019. (A) Total FA, major FA groups, and C:N ratio, and (B) EFAs. Error bars represent standard errors.

quality (Lovvorn et al. 2005; Tamelander et al. 2017; Rodil et al. 2020).

We find that the content of the major FA groups varies greatly between macroinvertebrate taxa but intraspecific variation over the season is low, with the exception of *Marenzelleria* and a spring-bloom related change in *Monoporeia*. The two amphipod species *Monoporeia* and *Pontoporeia* show similar FA content and cluster at trophic level two, indicating that they are primary consumers dependent on sedimenting organic material as food. The increase in EPA,  $\omega 3:\omega 6$  ratio and more negative  $\delta^{13}$ C for *Monoporeia* in April suggests rapid assimilation of spring-bloom material, which is supported by both field and experimental studies (Hill et al. 1992; Van de Bund et al. 2001; Dalsgaard et al. 2003; Lahdes et al. 2010). The lack of a phytoplankton-related FA signal in the amphipod *Pontoporeia* is likely related to its



**Fig. 3.** (A) Seasonal averaged content of PUFAs (including EFAs and their precursors ALA and LIN). The number reports the  $\omega 3:\omega 6$  PUFA ratio. (B) NMDS clustering of benthic macroinvertebrate species, sediment and seston based on the proportion of individual FAs over the whole sampling period. Polygons represent connections between the outermost scores in ordination of FAs in relation to each species, sediment and seston. Black points are individual FAs, with open points representing SAFAs; dominating FAs are labeled. The closer a species polygon is to a FA (point), the higher content of that particular FA is present in the species represented by the polygon.



**Fig. 4.** Biplot of  $\delta^{13}$ C and  $\delta^{15}$ N SI values for benthic macroinvertebrates, seston, and sediment in the Baltic Sea from July 2018 to April 2019. Error bars represent standard deviations.

deeper foraging strategy compared to *Monoporeia*, which may cause a competitive disadvantage for freshly deposited OM for this species (Hill and Elmgren 1987).

Our results show for the first time that *Marenzelleria*, a wide-spread ploychaete in the Northern Hemisphere and a successful invasive species in the Baltic Sea has the highest EFA content among the studied species. In addition, its low C:N ratio, indicative of high nutritional value for predators is in line with previous studies (Karlson et al. 2015). High levels

of EFA are surprising because Marenzelleria is adapted to forage in deeper sediment layers and likely has a higher tolerance for degraded and nutritionally poor food than other benthic macrofauna (Renz and Forster 2013). This is partially reflected in their comparably high levels of BAFAs and low ω3:ω6 ratios, indicating ingestion of high levels of FAs from terrestrial origin of low quality (Lau et al. 2014; Taipale et al. 2014). However, Marenzelleria was the only species showing a relationship with the diatom FA biomarker 16:1@-7 (palmitoleic acid; Dalsgaard et al. 2003) when spring bloom had settled in April (p < 0.01; Supporting Information Table S11). In addition, linear mixed effects regression between the two precursors 18:3 $\omega$ -3 (ALA) and 18:2 $\omega$ -6 (LIN), and their EFA end-products DHA and ARA (p < 0.001; Supporting Information Table S12), respectively, indicate the possibility of this species to elongate and desaturate EFAs from precursors. This suggests that Marenzelleria not only consumes old degraded and fresh sedimenting organic material but also has the possibility to produce EFA de novo. These observations are supported by a taxonomically widespread survey, indicating that polychaetes and other marine invertebrates have active desaturase genes to elongate and introduce double bonds from corresponding precursors (Kabeya et al. 2020).

In addition, the relatively high  $\delta^{15}$ N value and a DHA: EPA ratio of roughly 1 : 1 in *Marenzelleria* suggest an omnivorous feeding mode, which is contrary to the assumption of being a primary consumer (Dauer et al. 1981; Dauer 2000) and suggest possible trophic upgrading by the bacteria-based food chain that eventually provide EFA to *Marenzelleria*. High  $\delta^{15}$ N values, however, may also result from material ingestion from deeper sediment subjected to anaerobic ammonium oxidation

(anammox) that removes the lighter isotope (Brunner et al. 2013). Marenzelleria also showed a steady build-up of all major FA groups, including total FAs from summer toward late autumn (i.e., November), followed by a sharp drop for all FA groups between November and March, which is likely due to build-up of gametes, rich in FAs and spawning in late autumn (Bochert et al. 1997; Sikorski and Bick 2004; Kauppi et al. 2018). Overall, our findings strengthen previous experimental studies of Marenzelleria's competitive advantage compared to amphipods (Kotta and Ólafsson 2003) and its seasonal fluctuations in fresh adaptation to OM sedimentation.

Unlike Marenzelleria, the clam Macoma is low in FA content but nevertheless has a high C:N ratio, suggesting that this clam species has nutritional value for predators, including other lipid classes and glycogen (Graf et al. 1982; Wenne and Polak 1989). Low FA content in Macoma is in line with other studies showing that they accumulate low FA from primary production (Sun et al. 2009). This clam species has, however, an overall high w3:w6 ratio throughout the sampling period owing to relatively high levels of DHA, which is likely related to its ability to utilize EFAs from fresh pelagic OM (Lessin et al. 2019) and its selective assimilation of DHA. Surprisingly, the priapulid Halicryptus had similarly low levels of total FAs, including EFAs, despite its predatory feeding mode and expected FA increase with trophic level. Similar FA profiles in these two species are probably attributed to their tolerance for extended periods of hypoxia/anoxia conditions as anoxic metabolism uses glycogen as energy source in invertebrates, thus perhaps lowering the requirement for lipid reserves (Hochachka et al. 1973).

High interspecific variation in FA content of soft bottom benthic fauna as shown here is in line with observations for zooplankton and lake benthic macroinvertebrate species (Lau et al. 2012, 2021; Vesterinen et al. 2020). Similarly, the magnitude of seasonal fluctuations in FAs, especially EFAs are often taxa dependent among aquatic consumers (Goedkoop et al. 2000; Ravet et al. 2010; Lau et al. 2012). However, FA profiles of aquatic primary consumers may also be uncoupled to phytoplankton production, but rather vary with changing water temperature (McMeans et al. 2015; Lau et al. 2021), life history strategies related to reproduction, as shown for Marenzelleria, quality upgrading (Klein Breteler et al. 1999) or physiological FA regulation in anoxic environments, as assumed for Macoma and Halicryptus. The FA content of the benthic fauna investigated here did not respond to the spring inputs of OM. Although we can expect that sufficient time for settling of the spring bloom had passed (Rodil et al. 2020), making fresh OM available for the fauna as indicated by the increase in sediment EFAs in April. The lack of a response to spring bloom inputs is in contrast to previous studies, showing that total lipids, glycogen and FA percentages of some benthic animals respond to deposited spring OM, although the signal is usually short-lived (Graf et al. 1982;

Lehtonen 1996; Lahdes et al. 2010). However, our study agrees with Jardine et al. (Jardine et al. 2020), emphasizing that metabolic transformation, integration, and biosynthesis of certain FAs could mask the immediate dietary intake. Stable FA content of zooplankton and benthic species over the season suggest that these taxa have internal mechanisms to synthesize EFAs when supply is low or depend on food resources with low seasonal variation in biochemical composition. Moreover, low variation in abiotic conditions characteristic of deep soft bottom environments (Janas et al. 2007) may cause low turnover rates of FAs, especially in benthic consumers with low somatic growth rates and low lipid metabolism at low temperature. This suggests that FA composition of benthic macrofauna and the transfer of EFAs from primary producers to consumers are affected by different mechanisms related to a multitude of drivers (Goedkoop et al. 2000; Ravet et al. 2010; Lau et al. 2012, 2021; McMeans et al. 2015). Our results indicate that benthic organisms from multiple trophic levels and feeding guilds have evolved strategies to adapt to the low nutritional value provided by sediment OM, including the capability to biosynthesize EFA (Kabeya et al. 2018).

Several ecosystems experience decline in fisheries production, which is often linked to reduced dietary conversion of EFAs related to increasing cyanobacteria (Schmidt et al. 2020). We find that phytoplankton species composition and the quality of fresh settling OM does not directly translate into essential biomolecule composition for benthic animals, which are an important link to higher trophic levels. Our findings contradict the assumption that spring bloom sedimentation enhances nutritional value of FAs in benthic consumers, while summer blooming cyanobacteria settling is of reduced qualitative value and that this might have consequences for higher trophic levels. This indicates that the transfer of essential biomolecules from primary producer to upper trophic levels is more complex and not directly related to the taxonomic composition and content of phytoplankton. However, long-term increase in cvanobacteria biomass and reduced input of spring bloom material may affect consumer fecundity and growth rates and consequently community composition by favoring species best adapted to low-quality OM, rather than changing the FA profile of consumers.

#### References

- Arts, M. T., R. G. Ackman, and B. J. Holub. 2001. "Essential fatty acids" in aquatic ecosystems: A crucial link between diet and human health and evolution. Can. J. Fish. Aquat. Sci. 58: 122–137. https://doi.org/10.1139/cjfas-58-1-122
- Benke, A. C., and A. D. Huryn. 2010. Benthic invertebrate production—Facilitating answers to ecological riddles in freshwater ecosystems. J. N. Am. Benthol. Soc. 29: 264–285.
- Bochert, R., A. Bick, M. L. Zettler, and E.-A. Arndt. 1997. *Marenzelleria viridis* (Verrill, 1873) (Polychaeta: Spionidae), an invader in the benthic community in Baltic coastal

inlets—Investigation of reproduction, p. 131–139. *In Proceedings of the 13<sup>th</sup> symposium of the Baltic Marine Biologists*. Institute of Aquatic Ecology, University of Latvia.

- Brett, M. T., and D. C. Müller-Navarra. 1997. The role of highly unsaturated fatty acids in aquatic foodweb processes. Freshw. Biol. **38**: 483–499.
- Brunner, B., and others. 2013. Nitrogen isotope effects induced by anammox bacteria. Proc. Natl. Acad. Sci. U.S.A. **110**: 18994–18999.
- Dalsgaard, J., M. St John, G. Kattner, D. Müller-Navarra, and W. Hagen. 2003. Fatty acid trophic markers in pelagic marine environment. Adv. Mar. Biol. 46: 225–340.
- Dauer, D. M. 2000. Functional morphology and feeding behavior of *Spio setosa* (Polychaeta: Spionidae). Bull. Mar. Sci. 67: 269–275.
- Dauer, D. M., C. A. Maybury, and R. M. Ewing. 1981. Feeding behavior and general ecology of several spionid polychaetes from the Chesapeake Bay. J. Exp. Mar. Biol. Ecol. 54: 21–38.
- Galloway, A. W. E., and M. Winder. 2015. Partitioning the relative importance of phylogeny and environmental conditions on phytoplankton fatty acids. PLoS One **10**: 1–23.
- Goedkoop, W., L. Sonesten, G. Ahlgren, and M. Boberg. 2000. Fatty acids in profundal benthic invertebrates and their major food resources in Lake Erken, Sweden: Seasonal variation and trophic indications. Can. J. Fish. Aquat. Sci. **57**: 2267–2279.
- Graf, G., W. Bengtsson, U. Diesner, R. Schulz, and H. Theede. 1982. Benthic response to sedimentation of a spring phytoplankton bloom: Process and budget. Mar. Biol. **67**: 201–208.
- Griffiths, J. R., and others. 2017. The importance of benthicpelagic coupling for marine ecosystem functioning in a changing world. Glob. Chang. Biol. **23**: 2179–2196.
- Hill, C., and R. Elmgren. 1987. Vertical distribution in the sediment in the co-occurring benthic amphipods *Pontoporeia affinis* and *P. femorata*. Oikos **49**: 221.
- Hill, C., M. A. Quigley, J. F. Cavaletto, and W. Gordon. 1992. Seasonal changes in lipid content and composition in the benthic amphipods *Monoporeia affinis* and *Pontoporeia femorata*. Limnol. Oceanogr. **37**: 1280–1289.
- Hjerne, O., S. Hajdu, U. Larsson, A. Downing, and M. Winder. 2019. Climate driven changes in timing, composition and size of the Baltic Sea phytoplankton spring bloom. Front. Mar. Sci. 6: 1–15.
- Hochachka, P. W., J. Fields, and T. Mustafa. 1973. Animal life without oxygen: Basic biochemical mechanisms. Integr. Comp. Biol. 13: 543–555.
- Janas, U., E. Nowodworska, and I. Bezdzietny. 2007. Fitness and chemical composition of the Baltic clam *Macoma balthica* (Linnaeus, 1758) from sulphidic habitats in the Gulf of Gdańsk (Southern Baltic). Thermochim. Acta **458**: 112–117.
- Jardine, T. D., A. W. E. Galloway, and M. J. Kainz. 2020. Unlocking the power of fatty acids as dietary tracers and metabolic signals in fishes and aquatic invertebrates. Philos. Trans. R. Soc. B Biol. Sci. **375**: 20190639.

- Kabeya, N., M. M. Fonseca, D. E. K. Ferrier, J. C. Navarro, L. K. Bay, D. S. Francis, D. R. Tocher, L. F. C. Castro, and Ó. Monroig. 2018. Genes for de novo biosynthesis of omega-3 polyunsaturated fatty acids are widespread in animals. Sci. Adv. 4: 1–9.
- Kabeya, N., and others. 2020. Unique fatty acid desaturase capacities uncovered in *Hediste diversicolor* illustrate the roles of aquatic invertebrates in trophic upgrading. Philos. Trans. R. Soc. B Biol. Sci. **375**: 20190654.
- Karlson, A. M. L., E. Gorokhova, and R. Elmgren. 2015. Do deposit-feeders compete? Isotopic niche analysis of an invasion in a species-poor system. Sci. Rep. **5**: 9715.
- Kauppi, L., A. Norkko, and J. Norkko. 2018. Seasonal population dynamics of the invasive polychaete genus *Marenzelleria* spp. in contrasting soft-sediment habitats. J. Sea Res. **131**: 46–60.
- Klein Breteler, W. C. M., N. Schogt, M. Baas, S. Schouten, and G. W. Kraay. 1999. Trophic upgrading of food quality by protozoans enhancing copepod growth: Role of essential lipids. Mar. Biol. **135**: 191–198.
- Kortsch, S., R. Primicerio, M. Fossheim, A. V. Dolgov, and M. Aschan. 2015. Climate change alters the structure of arctic marine food webs due to poleward shifts of boreal generalists. Proc. R Soc. B Biol. Sci. 282: 20151546.
- Kotta, J., and E. Ólafsson. 2003. Competition for food between the introduced polychaete *Marenzelleria viridis* (Verrill) and the native amphipod *Monoporeia affinis* Lindström in the Baltic Sea. J. Sea Res. **50**: 27–35.
- Lahdes, E. O., T. Farkas, and K. K. Lehtonen. 2010. Phospholipid characteristics and neutral lipid fatty acid composition related to temperature and nutritional conditions in ecologically important amphipod species from the northern Baltic Sea. J. Exp. Mar. Biol. Ecol. **390**: 125–133.
- Lau, D. C. P., T. Vrede, J. Pickova, and W. Goedkoop. 2012. Fatty acid composition of consumers in boreal lakes— Variation across species, space and time. Freshw. Biol. **57**: 24–38.
- Lau, D. C. P., I. Sundh, T. Vrede, J. Pickova, and W. Goedkoop. 2014. Autochthonous resources are the main driver of consumer production in dystrophic boreal lakes. Ecology **95**: 1506–1519.
- Lau, D. C. P., A. Jonsson, P. D. F. Isles, I. F. Creed, and A. Bergström. 2021. Lowered nutritional quality of plankton caused by global environmental changes. Glob. Chang. Biol. 27: 6294–6306.
- Lehtonen, K. 1996. Ecophysiology of the benthic amphipod *Monoporeia affinis* in an open-sea area of the northern Baltic Sea: Seasonal variations in body composition, with bioenergetic considerations. Mar. Ecol. Prog. Ser. **143**: 87–98.
- Lehtonen, K. K., and A. B. Andersin. 1998. Population dynamics, response to sedimentation and role in benthic metabolism of the amphipod *Monoporeia affinis* in an open-sea area of the northern Baltic Sea. Mar. Ecol. Prog. Ser. **168**: 71–85.

23782242, 2023, 5, Downloaded from https

onlinelibrary.wiley.com/doi/10.1002/lol2.10333 by Swedish University Of Agricultural Sciences, Wiley Online Library on [11/10/2023]. See the Terms

and Conditions

· (https:

on Wiley Online

ibrary for rules of use; OA articles

the applicable Creative Commons License

- Lessin, G., J. Bruggeman, C. L. McNeill, and S. Widdicombe. 2019. Time scales of benthic macrofaunal response to pelagic production differ between major feeding groups. Front. Mar. Sci. **6**: 1–12.
- Lovvorn, J. R., L. W. Cooper, M. L. Brooks, C. C. De Ruyck, J. K. Bump, and J. M. Grebmeier. 2005. Organic matter pathways to zooplankton and benthos under pack ice in late winter and open water in late summer in the northcentral Bering Sea. Mar. Ecol. Prog. Ser. **291**: 135–150.
- Lutz, M. J., K. Caldeira, R. B. Dunbar, and M. J. Behrenfeld. 2007. Seasonal rhythms of net primary production and particulate organic carbon flux to depth describe the efficiency of biological pump in the global ocean. J. Geophys. Res. Oceans **112**: C10011.
- Martinez Arbizu, P. 2020. pairwiseAdonis: Pairwise multilevel comparison using adonis. R package version 0.4.
- McMeans, B. C., A. M. Koussoroplis, and M. J. Kainz. 2015. Effects of seasonal seston and temperature changes on lake zooplankton fatty acids. Limnol. Oceanogr. **60**: 573–583.
- Monroig, Ó., D. R. Tocher, and J. C. Navarro. 2013. Biosynthesis of polyunsaturated fatty acids in marine invertebrates: Recent advances in molecular mechanisms. Mar. Drugs **11**: 3998–4018.
- Oksanen, A. J., and others. (2020) Vegan: Community Ecology Package. R package Version 2.5-7.
- Petersen, G. H., and M. A. Curtis. 1980. Differences in energy flow through major components of subarctic, temperate and tropical marine shelf ecosystems. Dana **1**: 53–64.
- Pinheiro, J., and others. 2006. Linear, generalized linear, and nonlinear mixed effects models. *In Pharmacometrics: The science of quantitative pharmacology* (pp.103–135). Wiley Online Books.
- Ravet, R. L., M. T. Brett, and G. B. Arhonditsis. 2010. The effects of seston lipids on zooplankton fatty acid composition in Lake Washington, Washington, USA. Ecology **91**: 180–190.
- Renz, J. R., and S. Forster. 2013. Are similar worms different? A comparative tracer study on bioturbation in the three sibling species *Marenzelleria arctia*, *M. viridis*, and *M. neglecta* from the Baltic Sea. Limnol. Oceanogr. **58**: 2046–2058.
- Rodil, I. F., P. Lucena-Moya, T. Tamelander, J. Norkko, and A. Norkko. 2020. Seasonal variability in benthic–pelagic coupling: Quantifying organic matter inputs to the seafloor and benthic macrofauna using a multi-marker approach. Front. Mar. Sci. **7**: 1–18.
- Schmidt, K., and others. 2020. Increasing picocyanobacteria success in shelf waters contributes to long-term food web degradation. Glob. Chang. Biol. **26**: 5574–5587.
- Sikorski, A. V., and A. Bick. 2004. Revision of *Marenzelleria mesnil*, 1896 (Spionidae, Polychaeta). Sarsia **89**: 253–275.
- Simis, S. G. H., P. Ylöstalo, K. Y. Kallio, K. Spilling, and T. Kutser. 2017. Contrasting seasonality in optical biogeochemical properties of the Baltic Sea. PLoS One **12**: e0173357.
- Spilling, K., A. Kremp, R. Klais, K. Olli, and T. Tamminen. 2014. Spring bloom community change modifies carbon

pathways and C: N: P: Chl a stoichiometry of coastal material fluxes. Biogeosciences **11**: 7275–7289.

- Sun, M. Y., L. M. Clough, M. L. Carroll, J. Dai, W. G. Ambrose, and G. R. Lopez. 2009. Different responses of two common Arctic macrobenthic species (*Macoma balthica* and *Monoporeia affinis*) to phytoplankton and ice algae: Will climate change impacts be species specific? J. Exp. Mar. Biol. Ecol. **376**: 110–121.
- Svedäng, H., O. Savchuk, A. Villnäs, A. Norkko, B. G. Gustafsson, S. A. Wikström, and C. Humborg. 2022. Rethinking the "ecological envelope" of Eastern Baltic cod (*Gadus morhua*): Conditions for productivity, reproduction, and feeding over time. ICES J. Mar. Sci. **79**: 689–708.
- Taipale, S. J., M. J. Kainz, and M. T. Brett. 2014. A low  $\omega$ -3: $\omega$ -6 ratio in *Daphnia* indicates terrestrial resource utilization and poor nutritional condition. J. Plankton Res. **37**: 596–610.
- Taipale, S. J., K. Vuorio, U. Strandberg, K. K. Kahilainen, M. Järvinen, M. Hiltunen, E. Peltomaa, and P. Kankaala. 2016. Lake eutrophication and brownification downgrade availability and transfer of essential fatty acids for human consumption. Environ. Int. **96**: 156–166.
- Tamelander, T., K. Spilling, and M. Winder. 2017. Organic matter export to the seafloor in the Baltic Sea: Drivers of change and future projections. Ambio **46**: 842–851.
- Tiselius, P., B. W. Hansen, and D. Calliari. 2012. Fatty acid transformation in zooplankton: From seston to benthos. Mar. Ecol. Prog. Ser. **446**: 131–144.
- Van de Bund, W. J., E. Ólafsson, H. Modig, and R. Elmgren. 2001. Effects of the coexisting Baltic amphipods *Monoporeia affinis* and *Pontoporeia femorata* on the fate of a simulated spring diatom bloom. Mar. Ecol. Prog. Ser. **212**: 107–115.
- Vesterinen, J., and others. 2020. Nutritional quality of littoral macroinvertebrates and pelagic zooplankton in subarctic lakes. Limnol. Oceanogr. **66**: S81–S97.
- Wenne, R., and L. Polak. 1989. Lipid composition and storage in the tissues of the bivalve, *Macoma balthica*. Biochem. Syst. Ecol. 17: 583–587.
- Winder, M., and J. E. Cloern. 2010. The annual cycles of phytoplankton biomass. Philos. Trans. Roy. Soc. B Biol. Sci. 365: 3215–3226.

#### Acknowledgments

The authors want to thank the staff at the Askö field station for their support in sample collection and Dr. Martin Kainz for valuable comments on an earlier version of the manuscript. The authors acknowledge the support from the Swedish Metabolomics Centre by the Science for Life Laboratory (SciLifeLab). This project was funded by the Swedish Research Council Formas (Grant 2015-1320 to MW).

Submitted 13 October 2022 Revised 13 April 2023 Accepted 07 May 2023